

Evaluation of non-invasive biomarkers of kidney allograft rejection in a prospective multicenter unselected cohort study (EU-TRAIN)

Valentin Goutaudier, Richard Danger, Rusan Ali Catar, Maud Racapé, Aurélie Philippe, Michelle Elias, Marc Raynaud, Olivier Aubert, Didier Bouton, François Girardin, et al.

▶ To cite this version:

Valentin Goutaudier, Richard Danger, Rusan Ali Catar, Maud Racapé, Aurélie Philippe, et al.. Evaluation of non-invasive biomarkers of kidney allograft rejection in a prospective multicenter unselected cohort study (EU-TRAIN). Kidney International, 2024, S0085-2538 (24), pp.00565-00569. 10.1016/j.kint.2024.07.027. inserm-04712412

HAL Id: inserm-04712412 https://inserm.hal.science/inserm-04712412v1

Submitted on 27 Sep 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Evaluation of non-invasive biomarkers of kidney allograft rejection in a prospective multicenter unselected cohort study (EU-TRAIN)

OPEN

Valentin Goutaudier^{1,2}, Richard Danger³, Rusan Ali Catar⁴, Maud Racapé¹, Aurélie Philippe^{4,5}, Michelle Elias⁶, Marc Raynaud¹, Olivier Aubert^{1,2}, Didier Bouton⁷, François Girardin⁸, Éric Vicaut⁹, Sarhan Yaiche¹⁰, Jacques Demotes¹⁰, Harald Heidecke¹¹, Jean-Luc Taupin¹², Christine Randoux-Lebrun¹³, Mohamad Zaidan¹⁴, Emmanuelle Papuchon³, Hoa Le Mai³, Thi-Van-Ha Nguyen³, Francesc Moreso¹⁵, Thierry Berney¹⁶, Jean Villard¹⁷, Christophe Legendre², Duska Dragun⁴, Vassilios Papalois^{18,19}, Luciano Potena^{18,20}, Magali Giral^{3,21}, Pierre-Antoine Gourraud^{3,22}, Sophie Brouard^{3,21}, Elena Crespo²³, Fabian Halleck⁴, Klemens Budde⁴, Oriol Bestard^{15,23}, Alexandre Loupy^{1,2,24} and Carmen Lefaucheur^{1,6,24}; on behalf of the EU-TRAIN Consortium

¹Université Paris Cité, INSERM U970, Paris Institute for Transplantation and Organ Regeneration, Paris, France; ²Department of Kidney Transplantation, Necker Hospital, Assistance Publique—Hôpitaux de Paris, Paris, France; ³Centre Hospitalier Universitaire (CHU) Nantes, Nantes University, INSERM, Center for Research in Transplantation and Translational Immunology, UMR 1064, ITUN, Nantes, France; ⁴Department of Nephrology and Internal Intensive Care Medicine, Charité Universitätsmedizin Berlin, Corporate Member of Freie Universität Berlin and Humboldt-Universität zu Berlin, and Berlin Institute of Health (BIH), Berlin, Germany; ⁵BIH Biomedical Innovation Academy, Berlin Institute of Health at Charité—Universitätsmedizin Berlin (BIH), Berlin, Germany; ⁶Department of Kidney Transplantation, Saint-Louis Hospital, Assistance Publique—Hôpitaux de Paris, Paris, France; ⁷DRCI Direction of Clinical Research and Innovation, Assistance Publique—Hôpitaux de Paris, Paris, France; ⁸Division of Clinical Pharmacology, Department of Medicine and Department of Laboratory Medicine and Pathology, Lausanne University Hospital, Faculty of Medicine, University of Lausanne, Lausanne, Switzerland; ⁹Clinical Trial Unit Hospital, Lariboisière Saint-Louis Assistance Publique-Hôpitaux de Paris (AP-HP), Paris Cité University, Paris, France; ¹⁰ECRIN European Clinical Research Infrastructure Network, Paris, France; ¹¹CellTrend, Luckenwalde, Germany; ¹²Immunology and Histocompatibility Laboratory, Medical Biology Department, Saint-Louis Hospital, Paris, France; ¹³Department of Nephrology, Bichat Hospital, Assistance Publique—Hôpitaux de Paris, Paris, France; ¹⁴Department of Nephrology and Transplantation, Kremlin-Bicêtre Hospital, Assistance Publique—Hôpitaux de Paris, Le Kremlin-Bicêtre, France; ¹⁵Nephrology and Kidney Transplant Department, Vall d'Hebron University Hospital, Barcelona, Spain; ¹⁶Division of Transplantation, Department of Surgery, University of Geneva Hospitals, Geneva, Switzerland; ¹⁷Department of Immunology and Allergy and Department of Laboratory Medicine, Geneva University Hospital, Geneva, Switzerland; ¹⁸European Society for Organ Transplantation (ESOT); ¹⁹Imperial College Renal and Transplant Centre, Hammersmith Hospital, Imperial College Healthcare NHS Trust, London, UK; ²⁰Cardiology Unit, IRCCS Azienda Ospedaliero-Universitaria di Bologna, Bologna, Italy; ²¹Centre d'Investigation Clinique en Biothérapie, Centre de Ressources Biologiques (CRB), Centre Hospitalier Universitaire (CHU) Nantes, Nantes, France; ²²Nantes Université, Centre Hospitalier Universitaire (CHU) Nantes, Pôle Hospitalo-Universitaire 11: Santé Publique, Clinique des données, INSERM, Centre d'Investigation Clinique (CIC) 1413, Nantes, France; and ²³Translational Nephrology and Kidney Transplant Research Laboratory, Vall d'Hebron Research Institute (VHIR), Barcelona, Spain

Non-invasive biomarkers are promising tools for improving kidney allograft rejection monitoring, but their clinical adoption requires more evidence in specifically designed studies. To address this unmet need, we designed the EU-TRAIN study, a large prospective multicentric unselected cohort funded by the European Commission. Here, we included consecutive adult patients who received a kidney allograft in nine European transplant centers between November 2018 and June

²⁴AL and CLef are co-last authors.

Received 27 December 2023; revised 25 June 2024; accepted 26 July 2024

antibodies (DSA). The primary outcome was allograft rejection (antibody-mediated, T cell-mediated, or mixed) in the first year post-transplantation. Overall, 412 patients were included, with 812 biopsies paired with a blood sample. *CD4* gene expression was significantly associated with rejection, while circulating anti-HLA DSA had a significant association with allograft rejection and a strong association with antibody-mediated rejection. All other tested biomarkers, including *AKR1C3*, *CD3E*, *CD40*, *CD8A*, *CD9*, *CTLA4*, *ENTPD1*, *FOXP3*, *GZMB*, *ID3*, *IL7R*, *MS4A1*, *MZB1*, *POU2AF1*, *POU2F1*, *TCL1A*, *TLR4*, and *TRIB1*, as well as antibodies against angiotensin II type 1 receptor, endothelin 1 type A receptor, C3a and C5a receptors, did not show significant associations with

2020. We prospectively assessed gene expression levels of

19 blood messenger RNAs, four antibodies targeting non-

human leukocyte antigen (HLA) endothelial antigens,

together with circulating anti-HLA donor-specific

Correspondence: Carmen Lefaucheur, Kidney Transplant Department, Saint-Louis Hospital, 1 Avenue Claude Vellefeaux, 75010 Paris, France. E-mail: carmen.lefaucheur@aphp.fr; or Alexandre Loupy, Paris Institute for Transplantation and Organ Regeneration (PITOR), 56 rue Leblanc, Paris 75015, France. E-mail: alexandre.loupy@inserm.fr

allograft rejection. The blood messenger RNAs and non-HLA antibodies did not show an additional value beyond standard of care monitoring parameters and circulating anti-HLA DSA to predict allograft rejection in the first year post-transplantation. Thus, our results open avenues for specifically designed studies to demonstrate the clinical relevance and implementation of other candidate noninvasive biomarkers in kidney transplantation practice.

Kidney International (2024) ■, ■-■; https://doi.org/10.1016/ j.kint.2024.07.027

KEYWORDS: biomarkers; kidney transplantation; monitoring; rejection

Copyright © 2024, International Society of Nephrology. Published by Elsevier Inc. This is an open access article under the CC BY license (http:// creativecommons.org/licenses/by/4.0/).

Lay Summary

Optimal monitoring of kidney allograft rejection is essential to improve patient care but still relies on nonspecific biomarkers (e.g., serum creatinine and proteinuria), circulating anti-human leukocyte antigen (HLA) donor-specific antibodies (DSA) and surveillance biopsies. Blood gene expression profiling and non-HLA antibodies targeting endothelial antigens have shown associations with allograft rejection but remain to be validated in large, prospective, multicentric, and unselected cohorts. We designed the European TRAnsplantation and INnovation (EU-TRAIN) study (ClinicalTrials.gov, NCT03652402) to assess 19 blood mRNAs and 4 non-HLA antibodies in consecutive deeply phenotyped adult patients who received a kidney allograft between November 2018 and June 2020 in 9 European transplant institutions. All patients were prospectively followed from the time of transplantation until 1 year after transplantation. The EU-TRAIN blood biomarkers did not show an additional value beyond standard of care monitoring parameters and circulating anti-HLA DSA to predict allograft rejection. Hence, our results question their clinical utility for monitoring kidney allograft rejection in the first year after transplantation.

R ejection remains an important cause of kidney allograft loss, adversely impacting patient outcomes and constituting a significant public health problem.^{1,2} A precise, noninvasive rejection monitoring system is essential to obtain an accurate and timely detection and to provide optimal patient care.³ However, the field of kidney transplantation lacks robust clinically validated tools for immune monitoring, which still mainly relies on nonspecific biomarkers (e.g., serum creatinine and proteinuria), circulating anti–human leukocyte antigen (HLA) donor-specific antibodies (DSA), and surveillance biopsies.⁴ This represents a major issue for routine care, clinical trials, and improvement of transplant outcomes.⁴

Over the past 2 decades, several emerging noninvasive biomarkers have been developed to detect kidney allograft rejection, providing hope to improve the accuracy and timeliness of rejection diagnosis while reducing the need for invasive and costly biopsies.⁵⁻¹² Among these, blood gene expression profiling and antibodies targeting non-HLA endothelial antigens, also called non-HLA antibodies, have shown promising performances to detect clinical and subclinical rejection, either antibody-mediated (AMR) or T cellmediated (TCMR).^{7,13,14} Mechanistically driven gene expression levels of blood mRNAs have been associated with spontaneous tolerance and rejection.^{6,7,15–19} Most of these genes have been associated with immune regulation, including FOXP3, CTLA4, and CD40, and are expressed by circulating cells, including T, B, and natural killer cells, whose levels have been found altered in rejection events.²⁰ TRIB1 and TLR4 have been described as associated with chronic humoral rejection in case-control studies.¹⁵ Recently, 2 genes, TCL1A and AKR1C3, were associated and validated as biomarkers of subclinical rejection at 1 year after transplantation in 2 large cohorts. We also hypothesized that toleranceregulated genes may be modulated in case of immune tolerance breakdown, leading to rejection.^{6,7,16-19} Moreover, the association of non-HLA antibodies with auto- and alloimmunity and distinct allograft rejection patterns have also been demonstrated across multiple organ transplants.^{21–24}

However, their clinical adoption requires more evidence in specifically designed studies to correctly evaluate the added value beyond standard of care parameters, their context of use, and their transportability.^{25–29} There is a need for a robust assessment of these emerging blood biomarkers in a large, deeply phenotyped and unselected cohort of kidney transplant recipients to demonstrate their utility to inform clinical decision-making in routine practice.

To address this issue, we designed the European TRAnsplantation and INnovation (EU-TRAIN) study (ClinicalTrials. gov, NCT03652402), a large prospective longitudinal multicenter cohort study funded by the European Commission (grant agreement no. 754995), to investigate the clinical utility of 23 data-driven and mechanistically informed blood biomarkers (19 mRNA gene expression levels and 4 non-HLA antibodies) in monitoring kidney allograft rejection compared with the standard of care parameters. This study integrates and centralizes on a large scale a protocol-based assessment of clinical, biological, immunologic, and histologic variables using innovative technologies to constitute a multimodal cohort of unselected, extensively phenotyped adult kidney transplant recipients.

In this study, we aimed to (i) investigate the association of blood gene expression profiling and non-HLA antibodies, together with circulating anti-HLA DSA with kidney allograft rejection, and (ii) determine their additional predictive value as compared with parameters collected during the standard of care during the first year after transplantation.

METHODS

Ethics statement

The protocol of this study (ClinicalTrials.gov, NCT03652402) was approved by the institutional review board *Comité de Protection des Personnes du Sud-Ouest et Outre-Mer IV*

(registration approval number: 2018-A00733-52). All patients provided written informed consent to participate during the inclusion visit (i.e., at the time of transplantation). This research is governed by the Commission Nationale de l'informatique et des Libertés (French Data Protection Agency) "Reference Methodology for processing personal data used with the scope of health research" (amended MR-001). The study was conducted in accordance with the Declaration of Helsinki and the International Conference on Harmonization Guidelines for Good Clinical Practice, and the Declaration of Istanbul on Organ Trafficking and Transplant Tourism.

Study design and participants

The EU-TRAIN study is a large, unselected prospective multicenter cohort of deeply phenotyped adult kidney transplant recipients enrolled at the time of transplantation between November 2018 and June 2020 and followed during the first year after transplantation. This cohort is part of a collaborative network involving 9 European transplant centers (Paris-Saint-Louis, Paris-Necker, Nantes, Barcelona-Bellvitge, Barcelona-Vall d'Hebron, Berlin-Charité Mitte, Berlin-Charité Virchow, Geneva, and Paris-Kremlin-Bicêtre), 3 analytical platforms (Paris Cardiovascular Research Center INSERM U970, Center for Research in Transplantation and Translational Immunology UMR 1064 at Nantes University Hospital, and Medical Biology Department of Saint-Louis Hospital), and 1 industrial partner (CellTrend, Luckenwalde, Germany). Detailed inclusion and exclusion criteria are provided in the Supplementary Methods.

Patient recruitment started on the day of the kidney transplantation (baseline visit). Follow-up visits were performed per protocol at 3 months (M3) and 12 months (M12), and additional visits were conducted when a biopsy was clinically indicated (deterioration of kidney function and detection or rise in proteinuria).

We followed the STROBE (Strengthening the Reporting of Observational Studies in Epidemiology) statement checklist for the report of observational cohort studies.³⁰ We adhered to the STARD (Standards for Reporting of Diagnostic Accuracy Studies) checklist to ensure complete and transparent reporting of study our methods and results for assessing diagnostic accuracy.³¹ Moreover, we adhered to the SAGER (Sex and Gender Equity Research) guidelines for reporting sex and gender.³² Sex was self-reported by participants. All these complete checklists are provided in the Supplementary Material. The overall study design is presented in Figure 1. Details regarding sample size calculation are provided in the Supplementary Methods.

Multidimensional data collection and procedures

Clinical data and laboratory measurements. At baseline visit, clinical and biological data were collected related to (i) recipient characteristics, (ii) donor characteristics, (iii) transplant procedure, (iv) clinical examination, and (v) blood laboratory tests. At follow-up visits (M3 and M12, clinically indicated), a standardized transplant assessment

was performed, comprising (i) clinical examination, (ii) immunosuppressive treatment, (iii) blood and urinary analyses for standard of care laboratory parameters, (iv) blood analyses for innovative biomarkers, and (v) kidney allograft biopsy for histologic analysis. To note, allograft instability was defined as a 20% increase in serum creatinine and/or a worsening or appearance of a urine protein-to-creatinine ratio ≥ 0.5 g/g at the time of the follow-up visit.

The complete list of all routine parameters assessed during the study is available in Supplementary Table S1. The detailed immunosuppressive protocol, as well as procedure for data management, quality control, and privacy protection, is provided in the Supplementary Methods.

Immunological phenotyping. Kidney transplant recipients were tested for the presence of circulating donor-specific anti-HLA-A, -B, -Cw, -DR, -DQB, and -DP antibodies at baseline and each follow-up visit with single antigen flow bead assays (One Lambda, Inc.). Beads with a normalized mean fluo-rescence intensity (MFI) of greater than 500 units were considered positive, and beads with a normalized MFI of greater than 1400 were considered clinically meaningful, as recommended by international guidelines.³³ Immunodo-minant DSA was defined as the DSA with the highest MFI. All the sera were centrally reviewed in the HLA Saint Louis Hospital Laboratory.

HLA typing of recipients and donors were respectively performed by DNA typing (Innolipa HLA typing kit; Innogenetics) or next-generation sequencing (NGS-go; GenDx) with medium- to high-resolution sequence-specific primer (Linkage Biosciences, One Lambda) according to local HLA laboratory practices.

Histologic and immunohistochemical phenotyping. Kidney allograft biopsies were routinely performed at each followup visit according to local centers' practice. One kidney core was paraffin-embedded and formalin-fixed for histologic analysis. C4d staining was performed by immunohistochemistry on paraffin-embedded tissue or by immunofluorescence on frozen tissue according to local practices. Biopsies were assessed by local pathologists blinded to biomarker data, according to the international and standardized Banff 2019 classification for kidney allograft rejection (detailed international Banff scoring system is provided in Supplementary Table S2).^{3,34}

Blood biomarkers. We measured 23 biomarkers (19 blood mRNA gene expression levels and 4 non-HLA antibodies) previously studied as potential biomarkers for monitoring kidney allograft rejection,^{6,7,13,22} using blood samples collected and processed at each follow-up. All analytic platforms were blinded to the outcome measure (allograft rejection).

We quantified 19 mechanistically driven blood mRNA gene expression biomarkers that were previously associated with spontaneous operational tolerance or rejection (*AKR1C3, CD3E, CD4, CD40, CD8A, CD9, CTLA4, ENTPD1, FOXP3, GZMB, ID3, IL7R, MS4A1, MZB1, POU2AF1, POU2F1, TCL1A, TLR4,* and *TRIB1*)⁶ using NanoString PlexSet Technology (NanoString Technologies). As previously

clinical investigation



Figure 1 | European TRAnsplantation and INnovation (EU-TRAIN) multimodal biomarker study design. This figure illustrates the recruitment and follow-up of kidney transplant recipients included in the EU-TRAIN study. At each follow-up visit, we performed multidimensional phenotyping, comprising standard of care assessment and noninvasive biomarker evaluation. anti-HLA DSA, anti-human leukocyte antigen donor-specific antibody; ECD, expanded criteria donor. Parts of the figure created with BioRender.com.

described, all normalized counts of blood mRNAs were log2 transformed for subsequent statistical analyses.⁷

We analyzed 4 non-HLA antibodies directed against endothelial targets and involved in auto- and alloimmunity (angiotensin II type 1 receptor [AT1R] antibodies, endothelin 1 receptor type A [ETAR] antibodies, complement C3a receptor [C3aR] antibodies, and complement C5a receptor [C5aR] antibodies) in purified IgG in cooperation with CellTrend with commercially available solid phase assays (One Lambda, Inc.). As previously described in the literature, we measured the antibody levels in U/ml.²² We used 2 different clinically meaningful positive thresholds for aAT1R antibodies (10 U/ml and 17 U/ml) and 1 positive threshold for ETAR antibodies (10 U/ml), which were associated with allograft rejection in prior studies.²¹

The detailed analytical and data processing procedures are provided in the Supplementary Methods.

Outcomes measures

The primary outcome measure of the study was kidney allograft rejection, assessed blindly to biomarker data at each follow-up visit and defined as either active AMR, chronic active AMR, acute TCMR, chronic active TCMR, or mixed rejection (AMR and TCMR) according to the international Banff 2019 classification.³ Following the Banff 2019 criteria,

borderline/suspicious for acute TCMR cases and equivocal/ suspicious for diagnosis of AMR cases were not categorized as rejection.³

The secondary outcomes were subtypes of rejection, that is, AMR (referred to as active AMR or chronic active AMR) and TCMR (referred to as acute TCMR or chronic active TCMR).

Statistical analyses

Sample size calculation. Assuming a targeted prevalence of the primary outcome (kidney allograft rejection) of 8%–10% in our population, 0.05 acceptable difference in apparent and adjusted R-squared, and 0.05 margin of error in estimation of intercept, we calculated that the minimal sample size required for model development was approximately 400 patients in our study.³⁵ Hence, we aimed to assess the eligibility of 550 kidney transplant recipients, considering a potential loss to follow-up of around 25%–30%.

Data transformation and standardization. The distribution of each continuous variable was systematically assessed before analyses. Normality assumptions were evaluated using graphical methods (histogram plots and quantile-quantile plots). In cases where variables exhibited non-normal distributions (e.g., urine protein-to-creatinine ratio), we applied appropriate transformations guided by the nature of data (e.g., natural logarithm transformation for urine protein-tocreatinine ratio).

Descriptive analyses of baseline characteristics. We described continuous variables using means and standard deviations or medians and interquartile ranges (IQR) if appropriate. Means, medians, and proportions were compared using the Student *t* test (or the Wilcoxon test if appropriate) or the χ^2 test (or the Fisher exact test if appropriate), respectively.

Univariate association of blood biomarkers with kidney allograft rejection and collinearity. As the primary outcome of the study is allograft rejection, defined using kidney biopsies and laboratory measurements, it was predefined in the protocol of the study that the main statistical analysis will include patients who had at least 1 kidney biopsy with a concomitant biology during the study period. Each biopsy was considered as an independent observation.

We assessed the univariate association of the candidate blood biomarkers at the time of biopsy with the primary outcome (odds ratio [OR]) using logistic regression. We measured their discrimination ability through the area under the receiver operator characteristic curve (ROC-AUC). Then, we defined thresholds, according to the literature for AT1R and ETAR antibodies^{21,24} and to the maximum of the Youden index (i.e., the optimal cutoff that balances sensitivity and specificity in the best possible way) for other biomarkers without previously published thresholds, to calculate their other individual diagnostic performances to detect kidney allograft rejection (sensitivity, specificity, positive predictive value, and negative predictive value).

Finally, we assessed the collinearity between the biomarkers using a correlation matrix, with statistical correlations calculated using the Spearman test.

Additional value of biomarkers for predicting allograft rejection. We investigated the additional value of the candidate blood biomarkers to predict the individual risk of allograft rejection, as compared with standard of care parameters. For this purpose, we developed and compared elastic net–based predictive diagnostic models of rejection using either solely routinely collected parameters (referred to as the standard of care model), solely blood biomarkers (referred to as the biomarker model), or by integrating routinely collected parameters and blood biomarkers (referred to as the integrative model) at the time of biopsy. We used elastic net regularized regressions because this type of regression performs a feature selection, which allows to get the optimal number of variables in the final model, based on their predictive performances and considering their potential collinearity.

The dataset was randomly divided into train (65%) and test (35%) sets to respectively develop and internally validate the models. To avoid overfitting, combinations of hyperparameters that control the strength and type of applied regularization (alpha "mixing parameter" and lambda "regularization parameter") were optimized by robust 10-fold cross-validation during tuning the models.³⁶ In addition, the cross-validation process was repeated 5 times to minimize sampling bias. This type of robust internal validation in a targeted validation set that is representative of the intended population and setting may be sufficient to accurately validate the models and demonstrate their applicability in the population of interest.³⁷

We used the TRIPOD (Transparent Reporting of a multivariable Prediction Model for Individual Prognosis Or Diagnosis) statement for the reporting of the development and validation of the models. The performances of all predictive models were evaluated based on discrimination through calculation of the ROC-AUC. ROC-AUC were compared using the DeLong's test.³⁸ The dataset preparation and preprocessing steps, as well as the development and validation of the models, were performed with the *caret* package in R.^{39,40}

Handling of missing data. Univariate analyses were performed on complete cases datasets. For multivariable analyses, we performed a single imputation of missing values (if <30%, as recommended by international guidelines⁴¹) with a random forest algorithm using the *missForest* package in R.⁴² The maximum iteration was set to 10 times. Random forest algorithms handle mixed types of missing data, are adaptive to interactions and nonlinearity, and were found to consistently produce the lowest imputation error compared with other imputation methods when data are missing completely at random, as in the EU-TRAIN study due to the systematic data collection.^{43,44}

Software. All statistical analyses were carried out using R, version 4.2.2 (R Foundation for Statistical Computing) and Stata software, version 17.0 (StataCorp). *P* values below 0.05 were considered as significant. All the tests were 2 tailed.

Role of the funding sources

The funders of this study had no role in the study design, data collection, analysis, or interpretation of the manuscript.

RESULTS

Study population characteristics

Between November 2018 and June 2020, 412 kidney transplant recipients with at least 1 transplant biopsy with a concomitant biology were prospectively included in the main analysis of the study and followed during a median time of 12.12 months (IQR, 10.79–12.91 months) after transplantation (Figure 2).

The main characteristics of the patients at the time of transplantation are provided in Table 1. The mean recipient age was 53.4 (\pm 14.4) years, 264 (64.1%) patients were male, and 325 (80.6%) patients received transplants from deceased donors. The mean donor age was 55.0 (\pm 15.1) years. A total of 54 patients (13.4%) were retransplanted, and 82 (20.6%) had positive circulating anti-HLA DSA at the time of transplantation. Induction therapy predominantly consisted of rabbit antithymocyte globulin (58.3%).

At the time of biopsy, the mean estimated glomerular filtration rate was 48.4 (± 20.1) ml/min per 1.73 m², median urinary protein-to-creatinine ratio was 0.2 g/g (IQR, 0.1–0.4



Figure 2 | Flowchart of the European TRAnsplantation and INnovation (EU-TRAIN) study. M3, 3 months; M12, 12 months.

g/g), and positive circulating anti-HLA DSA were detected in 151 (19.6%) cases. The most frequently used immunosuppressive drugs were tacrolimus (88.3%), mycophenolate mofetil (82.6%), and steroids (93.6%) (Table 2). The mean numbers of allograft biopsies and EU-TRAIN blood biomarkers' measurements per patient were 1.97 (\pm 0.73) and 1.39 (\pm 0.74), respectively.

Detailed characteristics of participants stratified by center are provided in Table 3.

Kidney allograft phenotypes and outcome measure

A total of 812 kidney allograft biopsies were performed along with concomitant biological assessments between January 2019 and September 2021. The median time between kidney transplantation and allograft biopsy was 4.1 months (IQR, 3.0–12.1 months). The overall timeline of the biopsies is depicted in Supplementary Figure S1, with 317 (39.0%) biopsies performed at 3 months, 308 (37.9%) biopsies at 12 months, and 187 (23.0%) biopsies for clinical indication, mostly within the first 3 months after transplant (median time between kidney transplantation and clinically indicated allograft biopsies 2.04 months, IQR 0.56–6.57 months).

The cumulative incidence of rejection per patient was 8.1% (95% confidence interval [CI] = 5.7%-11%) at 1 year after transplantation. Among the 812 biopsies performed in the 412 patients included in the main analysis, rejection was diagnosed in 52, including active AMR (n = 18), chronic active AMR (n = 6), acute TCMR (n = 18), chronic active

TCMR (n = 7), and mixed rejection (n = 3) (Table 2). Among these 52 rejection cases, 18 (34.6%) were subclinical (i.e., detected on protocol biopsies with a stable kidney function and no significant proteinuria). In 49 biopsies, we observed other rejection-related diagnoses, comprising borderline/suspicious for acute TCMR (n = 27), equivocal/ suspicious for diagnosis of AMR (n = 9), chronic inactive AMR (n = 1), and C4d deposition without evidence of rejection (n = 12). Non-rejection-related diagnoses reflected the full spectrum of kidney transplant pathology, including minimal or no histopathologic findings (n = 513), calcineurin inhibitor toxicity (n = 29), acute tubular injury (n =35), thrombotic microangiopathy (n = 15), BK virus nephropathy (n = 16), recurrent or *de novo* glomerulonephritis (n = 32), isolated moderate to severe interstitial fibrosis and tubular atrophy (n = 60), and other diagnoses (n = 11). Detailed characteristics of biopsies are provided in Table 2. The corresponding international Banff scores are provided in Supplementary Table S3.

Univariate association between EU-TRAIN blood biomarkers and kidney allograft rejection

Gene expression levels of *CD4* were significantly associated with allograft rejection (OR = 0.49, 95% CI = 0.32–0.73, P < 0.001). All other blood mRNA gene expression levels were not significantly associated with allograft rejection (Figure 3). There was a multicollinearity between blood mRNA gene expression levels (Figure 4). Their diagnostic

Table 1 | Recipient, donor, and transplant characteristics of patients included in the EU-TRAIN biomarker cohort (N = 412)

Variables	n ^a	Values
Recipient characteristics		
Age, yr, mean (SD)	412	53.4 (14.4)
Sex, n (%)	412	
Female		148 (35.9)
Male		264 (64.1)
Cause of end-stage kidney disease, n (%)	412	
ADPKD		58 (14.1)
Diabetes		41 (10.0)
Vascular		75 (18.2)
Glomerulonephritis		128 (31.1)
Tubulointerstitial		31 (7.5)
Other		41 (10.0)
Unknown		38 (9.2)
Donor characteristics		
Age, yr, mean (SD)	398	55.0 (15.1)
Blood group, n (%)	411	
A		149 (36.3)
AB		20 (4.9)
В		47 (11.4)
0		195 (47.4)
Type, n (%)	403	
Living		78 (19.4)
Deceased standard criteria donor		166 (41.2)
Deceased expanded criteria donor		159 (39.5)
Transplant baseline characteristics		
Prior kidney transplantation, n (%)	402	54 (13.4)
Cold ischemia time, h, n (%)	398	12.8 (8.2)
Delayed graft function, n (%)	412	99 (24.0)
HLA A/B/DR mismatches, median [IQR]	411	4.0 [3.0-5.0]
ABO incompatible transplantation, n (%)	412	17 (4.1)
Anti-HLA DSA at the time of transplantation, n (%)	399	82 (20.6)
Induction therapy, n (%)	412	
No induction or baliximab		172 (41.7)
Thymoglobulin		240 (58.3)

ABO, ABO blood group; ADPKD, autosomal dominant polycystic kidney disease; anti-HLA DSA, anti-human leukocyte antigen donor-specific antibody; EU-TRAIN, European TRAnsplantation and INnovation; IQR, interquartile range.

^aNumber of observations with available data regarding each variable of interest.

performances for rejection during the first year after transplantation, including discriminative ability, were low to moderate (Table 4).

The median antibody levels at the time of biopsy were 9.32 U/ml (IQR, 7.58–11.69 U/ml) for AT1R, 9.43 U/ml (IQR, 7.84–11.72 U/ml) for ETAR, 5.56 U/ml (IQR, 3.59–8.62 U/ml) for C3aR, and 11.29 U/ml (IQR, 7.05–18.20 U/ml) for C5aR. We did not find significant associations between these antibody levels and allograft rejection (Figure 5). There was a significant multicollinearity between the non-HLA antibody levels (Figure 4). Their diagnostic performances for rejection were low (AT1R antibodies: ROC-AUC = 0.55, 95% CI = 0.46–0.65, ETAR antibodies: ROC-AUC = 0.56, 95% CI =

0.47-0.66, C3aR antibodies: ROC-AUC = 0.51, 95% CI = 0.42-0.61, C5aR antibodies: ROC-AUC = 0.56, 95% CI = 0.46-0.66; see Table 5 for other metrics).

Univariate association between standard of care parameters and kidney allograft rejection

Next, we evaluated the univariate association of routine parameters (complete list available in Supplementary Table S1) with allograft rejection in the EU-TRAIN study population. The history of rejection (OR = 4.96, 95% CI = 2.37-10.39, P < 0.001), ABO blood group incompatibility between the recipient and donor (OR = 2.79, 95% CI = 1.11-6.97, P =(OR) 0.03), induction therapy with antithymocyte globulin (OR = 0.36, 95% CI = 0.20–0.65, P = 0.001), allograft instability at the time of biopsy (OR = 7.49, 95% CI = 4.12-13.63, P < 0.001), estimated glomerular filtration rate at the time of biopsy (OR = 0.96, 95% CI = 0.95–0.98, P < 0.001), urine protein-to-creatinine ratio at the time of biopsy (OR = 1.41, 95% CI = 1.07–1.86, P = 0.016), and MFI of the immunodominant anti-HLA DSA at the time of biopsy (OR = 1.56, 95% CI = 1.03–2.38, P = 0.04) were significantly associated with kidney allograft rejection (Figure 6). Circulating anti-HLA DSA were also strongly associated with AMR (OR =3.20, 95% CI = 1.84–5.55, P < 0.001). The individual diagnostic performances of standard of care parameters are provided in Supplementary Table S4.

Additional predictive value of blood biomarkers compared with routine parameters

To assess the added value of the blood biomarkers to predict kidney allograft rejection beyond standard of care parameters, we developed and compared a model trained using solely standard of care parameters (referred to as the standard of care model), a model trained using solely blood biomarkers (referred to as the biomarker model), and a model trained using both standard of care parameters and blood biomarkers (referred to as the integrative model). Optimal combinations of predictors were selected in each model by elastic net regularized regressions. To note, 6 blood mRNA biomarkers (*AKR1C3, CTLA4, FOXP3, MZB1, POU2AF1,* and *TCL1A*) were excluded from these analyses because of a rate of missing values >30% mainly secondary to technical issues in routine monitoring.

Clinical predictors of allograft rejection in the standard of care model were allograft instability, MFI of the immunodominant anti-HLA DSA, history of rejection, induction therapy with antithymocyte globulin, delayed graft function, urine protein-to-creatinine ratio (log transformed), and estimated glomerular filtration rate (Supplementary Figure S2A). The discriminative ability of this model for allograft rejection was moderate (ROC-AUC = 0.76, 95% CI = 0.63-0.89; Figure 7).

Blood biomarkers at the time of biopsy included in the biomarker model had a low discriminative ability for allograft rejection (ROC-AUC = 0.60, 95% CI = 0.46-0.75; Figure 7).

clinical investigation

Table 2 | Histologic findings and biological parameters at the time of kidney allograft biopsies performed during the EU-TRAIN study (N = 812)

Variables	n ^a	Values
Biopsy characteristics		
Biopsy indication, n (%)	812	
Protocol M3		317 (39.0)
Protocol M12		308 (37.9)
Clinically indicated		187 (23.0)
Main diagnosis, n (%)	812	
Active AMR		18 (2.2)
Chronic active AMR		6 (0.7)
Chronic inactive AMR		1 (0.1)
Acute TCMR		18 (2.2)
Chronic active TCMR		7 (0.9)
Mixed rejection (AMR + TCMR)		3 (0.4)
Borderline/suspicious for acute TCMR		27 (3.3)
Equivocal/suspicious for diagnosis of AMR ^b		9 (1.1)
C4d without evidence of rejection		12 (1.5)
BK virus nephropathy		16 (2.0)
Glomerulonephritis (recurrent or <i>de novo</i>)		32 (3.9)
CNI toxicity without rejection		29 (3.6)
TMA without rejection nor CNI toxicity		15 (1.8)
Acute tubular injury without rejection		35 (4.3)
Isolated IFTA ≥ 2		60 (7.4)
Other diagnoses		11 (1.4)
Normal or minimal changes		513 (63.2)
Biology at the time of biopsy		
eGFR (MDRD formula, ml/min per 1.73 m ²), mean (SD)	790	48.4 (20.1)
Urine protein-to-creatinine ratio, g/g, median [IQR]	662	0.2 [0.1–0.4]
Anti-HLA DSA, n (%)	772	152 (19.7)
MFI of the immunodominant anti-HLA DSA, ^c n (%)	771	
<500		620 (80.4)
500-1400		93 (12.1)
>1400		58 (7.5)
Immunosuppressive therapy at the time of biopsy		
Steroids, n (%)	787	737 (93.6)
Antimetabolite therapy, n (%)	787	
Mycophenolate mofetil		650 (82.6)
Azathioprine		48 (6.1)
No		89 (11.3)
mTOR inhibitor therapy, n (%)	784	60 (7.7)
Calcineurin inhibitor therapy, n (%)	788	
Tacrolimus		696 (88.3)
Cyclosporine		44 (5.6)

(Continued)

Table 2 (Continued)

-		
/ariables	n ^a	Values
No		48 (6.1)
Belatacept, n (%)	785	47 (6.0)

AMR, antibody-mediated rejection; anti-HLA DSA, anti-human leukocyte antigen donor-specific antibody; C4d, complement component C4d; CNI, calcineurin inhibitor; DSA, donor-specific antibody; eGFR, estimated glomerular filtration rate; EU-TRAIN, European TRAnsplantation and INnovation; IFTA, interstitial fibrosis and tubular atrophy; IQR, interquartile range; MDRD, Modification of Diet in Renal Disease formula; MFI, mean fluorescence intensity; mTOR, mammalian target of rapamycin; TCMR, T cell-mediated rejection; TMA, thrombotic microangiopathy.

Clinical indications for performing a kidney allograft biopsy were deterioration of kidney function and/or detection or rise in proteinuria.

^aNumber of observations with available data regarding each variable of interest. ^bThe category "Equivocal/suspicious for diagnosis of AMR" refers to biopsies with lesions evocative of AMR but not fulfilling all the criteria for a diagnosis of AMR (moderate microvascular inflammation [g + ptc ≥ 2 in the absence of recurrent or *de novo* glomerulonephritis and at least g ≥ 1 in the presence of acute TCMR, borderline infiltrate or infection] without circulating DSA or a substitute [complement C4d staining or expression of validated molecular markers] and C4d-positive acute tubular injury without circulating DSA).

 $^{\rm Cl}{\rm Immunodominant}$ DSA refers to the class of DSA with the highest level of fluorescence signal during measurement.

Their individual predictive importance is provided in Supplementary Figure S2B.

Finally, none of the blood biomarkers at the time of biopsy were predictors of allograft rejection in the integrative model (Supplementary Figure S2C). Moreover, the discriminative ability of this model for allograft rejection was moderate (ROC-AUC = 0.78, 95% CI = 0.67–0.88; Figure 7) and similar to the one of the standard of care model (P = 0.45).

Sensitivity analyses

Finally, various sensitivity analyses were performed to further confirm our findings and test the added value of blood biomarkers in different clinical scenarios and subpopulations. In all these scenarios, we confirmed the absence of incremental predictive ability of the 13 well-expressed blood mRNAs and 4 non-HLA antibodies targeting endothelial antigens to detect allograft rejection. These scenarios included (i) stable or unstable patients (i.e., with a deterioration of kidney function and/or detection or rise in proteinuria at the time of biopsy), (ii) rejection subtypes (i.e., AMR and TCMR), (iii) timing of biomarker measurement when assessed in the first 3 months after transplantation or after 3 months after transplantation, (iv) type of immunosuppressive therapy (i.e., with or without calcineurin inhibitor), (v) center effect, and (vi) kidney allograft rejection defined according to the international Banff 2022 classification⁴⁵ (Table 6).

DISCUSSION

In this large, prospective, and unselected cohort of 412 deeply phenotyped adult kidney transplant recipients with a systematic assessment of biomarkers during the first year after transplantation, the 23 EU-TRAIN blood biomarkers showed multicollinearity and did not demonstrate an additional value beyond standard of care parameters and circulating anti-HLA DSA for predicting allograft rejection.

We studied 19 mechanistically driven blood gene expression biomarkers that we and others found associated

			EU-IRAIN's transplant centers								
Variable	N ^a	Overall $(N = 412)$	Paris-Saint- Louis (n = 119)	Paris-Necker (n = 117)	Nantes (n = 43)	Barcelona- Bellvitge (n = 61)	Charité-Mitte (n = 21)	Charité- Virchow (n= 12)	Paris-Kremlin- Bicêtre (n = 20)	Barcelona- Vall d'Hebron (n = 6)	Genève (n = 13)
Recipient characteristics											
Age, yr, mean (SD)	412	53.4 (14.4)	52.3 (15.1)	52.8 (15.2)	53.4 (14.6)	55.1 (13.4)	55.0 (11.7)	49.3 (15.0)	56.1 (11.9)	62.7 (10.9)	52.0 (13.5)
Sex, n (%)	412										
Female		148 (35.9)	44 (37.0)	41 (35.0)	19 (44.2)	18 (29.5)	11 (52.4)	2 (16.7)	9 (45.0)	1 (16.7)	3 (23.1)
Male		264 (64.1)	75 (63.0)	76 (65.0)	24 (55.8)	43 (70.5)	10 (47.6)	10 (83.3)	11 (55.0)	5 (83.3)	10 (76.9)
Cause of end-stage kidney disease, n (%)	412										
ADPKD		58 (14.1)	8 (6.7)	28 (23.9)	8 (18.6)	3 (4.9)	7 (33.3)	2 (16.7)	0 (0.0)	1 (16.7)	1 (7.7)
Diabetes		41 (10.0)	18 (15.1)	6 (5.1)	1 (2.3)	10 (16.4)	0 (0.0)	1 (8.3)	3 (15.0)	2 (33.3)	0 (0.0)
Vascular		75 (18.2)	28 (23.5)	17 (14.5)	6 (14.0)	7 (11.5)	4 (19.0)	0 (0.0)	9 (45.0)	1 (16.7)	3 (23.1)
Glomerulonephritis		128 (31.1)	36 (30.3)	36 (30.8)	12 (27.9)	18 (29.5)	7 (33.3)	7 (58.3)	4 (20.0)	2 (33.3)	6 (46.2)
Tubulointerstitial		31 (7.5)	8 (6.7)	7 (6.0)	7 (16.3)	6 (9.8)	1 (4.8)	0 (0.0)	2 (10.0)	0 (0.0)	0 (0.0)
Other		41 (10.0)	6 (5.0)	15 (12.8)	7 (16.3)	8 (13.1)	0 (0.0)	1 (8.3)	1 (5.0)	0 (0.0)	3 (23.1)
Unknown		38 (9.2)	15 (12.6)	8 (6.8)	2 (4.7)	9 (14.8)	2 (9.5)	1 (8.3)	1 (5.0)	0 (0.0)	0 (0.0)
Donor characteristics											
Age, yr, mean (SD)	398	55.0 (15.1)	53.1 (16.4)	56.9 (15.0)	54.7 (14.6)	55.6 (15.2)	52.5 (14.9)	53.0 (8.2)	54.9 (11.6)	59.0 (10.8)	56.1 (12.3)
Blood group, n (%)	411										
А		149 (36.3)	40 (33.6)	36 (30.8)	19 (44.2)	29 (47.5)	2 (10.0)	6 (50.0)	9 (45.0)	3 (50.0)	5 (38.5)
AB		20 (4.9)	7 (5.9)	6 (5.1)	0 (0.0)	4 (6.6)	0 (0.0)	2 (16.7)	0 (0.0)	1 (16.7)	0 (0.0)
В		47 (11.4)	14 (11.8)	11 (9.4)	3 (7.0)	8 (13.1)	5 (25.0)	2 (16.7)	4 (20.0)	0 (0.0)	0 (0.0)
0		195 (47.4)	58 (48.7)	64 (54.7)	21 (48.8)	20 (32.8)	13 (65.0)	2 (16.7)	7 (35.0)	2 (33.3)	8 (61.5)
Type, n (%)	403										
Living		78 (19.4)	6 (5.0)	38 (32.5)	8 (18.6)	4 (6.9)	6 (30.0)	6 (85.7)	3 (15.0)	0 (0.0)	7 (53.8)
Deceased SCD		166 (41.2)	61 (51.3)	36 (30.8)	18 (41.9)	29 (50.0)	8 (40.0)	1 (14.3)	9 (45.0)	2 (33.3)	2 (15.4)
Deceased ECD		159 (39.5)	52 (43.7)	43 (36.8)	17 (39.5)	25 (43.1)	6 (30.0)	0 (0.0)	8 (40.0)	4 (66.7)	4 (30.8)
Transplant baseline characteristics											
Prior kidney transplantation, n (%)	402	54 (13.4)	22 (18.8)	10 (8.6)	4 (9.8)	10 (16.4)	0 (0.0)	2 (16.7)	2 (10.0)	2 (33.3)	2 (15.4)

Table 3 | Recipient, donor, and transplant characteristics of patients per center included in the EU-TRAIN cohort

(Continued on following page)

			EU-TRAIN's transplant centers								
Variable	N ^a	Overall $(N = 412)$	Paris-Saint- Louis (n = 119)	Paris-Necker (n = 117)	Nantes (n = 43)	Barcelona- Bellvitge (n = 61)	Charité-Mitte (n = 21)	Charité- Virchow (n= 12)	Paris-Kremlin- Bicêtre (n = 20)	Barcelona- Vall d'Hebron (n = 6)	Genève (n = 13)
Cold ischemia time, h, mean (SD)	398	12.8 (8.2)	13.6 (4.7)	13.3 (10.5)	10.4 (8.3)	15.5 (8.7)	9.2 (5.8)	6.7 (6.0)	11.6 (6.5)	16.4 (6.0)	6.8 (7.5)
Delayed graft function, n (%)	412	99 (24.0)	37 (31.1)	16 (13.7)	3 (7.0)	23 (37.7)	6 (28.6)	4 (33.3)	6 (30.0)	1 (16.7)	3 (23.1)
HLA A/B/DR mismatches, median [IQR]	411	4.0 [3.0–5.0]	4.0 [3.0–5.0]	4.0 [3.0–5.0]	4.0 [3.0–5.0]	4.0 [3.0–5.0]	3.5 [2.8–5.0]	3.0 [2.8–5.0]	4.0 [3.0–5.0]	3.5 [2.3–4.0]	5.0 [3.0–5.0]
ABO incompatible transplantation, n (%)	412	17 (4.1)	1 (0.8)	5 (4.3)	1 (2.3)	1 (1.6)	0 (0.0)	4 (33.3)	2 (10.0)	1 (16.7)	2 (15.4)
Anti-HLA DSA at the time of transplantation, n (%)	399	82 (20.6)	43 (36.1)	29 (25.0)	2 (5.1)	6 (10.5)	0 (0.0)	0 (0.0)	1 (5.0)	0 (0.0)	1 (8.3)
Induction therapy, n (%)	412										
No induction or baliximab		172 (41.7)	2 (1.7)	79 (67.5)	1 (2.3)	36 (59.0)	21 (100.0)	12 (100.0)	9 (45.0)	2 (33.3)	10 (76.9)
Thymoglobulin		240 (58.3)	117 (98.3)	38 (32.5)	42 (97.7)	25 (41.0)	0 (0.0)	0 (0.0)	11 (55.0)	4 (66.7)	3 (23.1)

Table 3 | (Continued) Recipient, donor, and transplant characteristics of patients per center included in the EU-TRAIN cohort

ABO, ABO blood group; ADPKD, autosomal dominant polycystic kidney disease; anti-HLA DSA, anti-human leukocyte antigen donor-specific antibody; ECD, expanded criteria donor; EU-TRAIN, European TRAnsplantation and INnovation; IQR, interquartile range; SCD, standard criteria donor.

^aNumber of observations with available data regarding each variable of interest.



Figure 3 Association of blood mRNAs at the time of biopsy with kidney allograft rejection. Pirate plots displaying the univariate association between blood mRNA gene expression levels at the time of biopsy and kidney allograft rejection. The horizontal bars represent medians, the bean shapes represent smooth density curves showing the full data distribution, the bands represent Bayesian highest density intervals, and the black points represent individual raw data points. Normalized counts of blood mRNAs were log2-transformed, as previously described in the literature.

with immune response, spontaneous tolerance, DSA generation, and subclinical rejection in previous studies.^{6,7,15} In this unselected cohort reflecting natural disease prevalence, we demonstrate that they were not associated with allograft rejection after adjustment for standard of care parameters, including in a sensitivity analysis with only protocol biopsies (i.e., aiming to focus on subclinical rejection). Their added value combined in an integrative score with standard of care parameters remains to be further evaluated.⁷

We also assessed 4 non-HLA antibodies targeting endothelial antigens (AT1R, ETAR, C3aR receptor, and C5aR) involved in auto- and alloimmunity in multiple organ transplants.²¹ Among them, AT1R and ETAR antibodies showed promising performances to discriminate kidney allograft rejection, especially AMR, and facilitated immunologic risk stratification in prior retrospective studies.^{13,22} Here we show that none of them were associated with kidney allograft rejection in univariate analysis, including in a sensitivity analysis excluding TCMR and mixed rejection cases and focusing on AMR. Previous studies showed causal associations between AT1R antibody levels and rejection episodes in limited and specific scenarios.^{22,23} The EU-TRAIN study assessed different contexts of use and populations, and investigated these biomarkers in a systematic prospective monitoring with a dedicated study design. The current results are consistent with a recent exploratory analysis, which indicated no association between AT1R and ETAR antibody levels and allograft rejection.⁴⁶ In addition, they align with the Sensitization in Transplantation: Assessment of Risk (STAR) working group guidelines, which do not recommend routine post-transplant assessment of non-HLA antibodies because of the insufficient evidence regarding their clinical utility.⁴⁷ Moreover, we confirmed that a proper confrontation with standard of care parameters is challenging, albeit crucial to demonstrate the clinical utility of biomarkers to inform decision-making.²⁵ Indeed, we demonstrated in this study that a model based solely on 7 simple clinical routine parameters (allograft instability, MFI of the immunodominant DSA, previous episode of rejection, induction therapy with antithymocyte globulin, delayed graft function, urine proteinto-creatinine ratio, and estimated glomerular filtration rate), previously established as factors associated with rejection,^{48–51} resulted in a reasonable discriminative ability for allograft rejection (ROC-AUC = 0.76, 95% CI = 0.63-0.89). Among the 23 candidate EU-TRAIN blood biomarkers, none showed superior discriminative abilities, even combined in a model developed using solely biomarkers (ROC-AUC = 0.60, 95%CI = 0.46-0.75). To note, the high negative predictive values observed in this study were due to the low prevalence of rejection-which reflects our unselected design and the realworld evidence of improvement of rejection prevention with modern immunosuppressive therapies⁵²—and not to good diagnostic performances (negative predictive values are



Figure 4 Association of non-human leukocyte antigen (non-HLA) antibodies against endothelial targets at the time of biopsy with kidney allograft rejection. Pirate plots displaying the univariate association between levels of non-HLA antibodies against endothelial targets at the time of biopsy and kidney allograft rejection. The horizontal bars represent medians, the bean shapes represent smooth density curves showing the full data distribution, the bands represent Bayesian highest density intervals, and the black points represent individual raw data points. Non-HLA antibody levels were measured in U/ml. AT1R, angiotensin II receptor type 1; C3aR, complement C3a receptor; C5aR, complement C5a receptor; ETAR, endothelin 1 receptor type A.

inversely proportional to disease prevalence),⁵³ and thus should not be considered.

The robustness of our data results from (i) our unselected study design, which captured the real-world distribution of kidney allograft phenotypes; (ii) the confirmation of our findings in various exploratory sensitivity analyses for stable and unstable patients, subpopulations of AMR and TCMR, timing of biomarker measurement after transplantation, type of immunosuppressive therapy, across centers, and definition of kidney allograft rejection according to the international

Table 4 | Diagnostic performances of blood mRNA gene expression levels at the time of biopsy to detect kidney allograft rejection

Plaad mPNA biomarkars	Discriminative ability,	Sonsitivity (%)	Specificity (%)	Positive predictive	Negative predictive
			Specificity (%)		
AKR1C3	0.51 (0.39–0.63)	92.31	15.58	5.31	97.53
CD3E	0.55 (0.46–0.64)	80.00	37.27	7.00	96.93
CD4	0.68 (0.60-0.76)	77.14	52.95	8.82	97.52
CD40	0.52 (0.42–0.63)	54.55	57.78	6.79	95.75
CD8A	0.56 (0.47–0.66)	82.86	34.18	6.94	97.12
CD9	0.54 (0.44–0.63)	91.43	22.02	6.45	97.76
CTLA4	0.59 (0.49–0.69)	83.33	40.26	7.91	97.52
ENTPD1	0.54 (0.43-0.64)	62.86	48.66	6.71	95.71
FOXP3	0.53 (0.36–0.73)	88.89	35.16	6.35	98.46
GZMB	0.57 (0.46-0.67)	22.86	91.84	14.29	95.24
ID3	0.50 (0.40-0.59)	80.00	27.35	6.29	95.73
IL7R	0.53 (0.44-0.63)	68.57	43.34	6.67	95.90
MS4A1	0.58 (0.48-0.67)	81.25	37.52	6.77	97.29
MZB1	0.54 (0.43-0.66)	75.00	38.35	5.68	96.88
POU2AF1	0.50 (0.39–0.62)	19.35	90.91	12.00	94.62
POU2F1	0.52 (0.43-0.61)	85.71	27.73	6.52	97.06
TCL1A	0.56 (0.48-0.68)	43.33	75.09	8.61	96.07
TLR4	0.53 (0.42-0.63)	25.71	89.95	13.04	95.38
TRIB1	0.51 (0.42–0.60)	78.79	34.86	6.36	96.70

Cl, confidence interval; ROC-AUC, area under the receiver operating characteristic curve.

ARTICLE IN PRESS



Figure 5 | **Multicollinearity between EU-TRAIN, European TRAnsplantation and INnovation blood biomarkers.** This correlation matrix shows the relationship between the values of blood biomarkers. Only statistically significant correlations are represented in each box. Positive correlations are in blue and negative correlations are in red. The diameter of each correlation circle is proportional to the strength of the correlation. The clusters of correlations are framed in black. AT1R_Ab, antibodies against angiotensin II receptor type 1; C3aR_Ab, antibodies against complement C3a receptor; C5aR_Ab, antibodies against complement C5a receptor; ETAR_Ab, antibodies against endothelin 1 receptor type 1.

Table 5 | Diagnostic performances of non-HLA antibodies against endothelial targets at biopsy to detect kidney allograft rejection

Non-HLA antibodies against	Discriminative ability, ROC-AUC (95% CI)	Sensitivity (%)	Specificity (%)	Positive predictive value (%)	Negative predictive value (%)
AT1R	0.55 (0.46–0.65)				
Low threshold (10 U/ml)	-	38.46	58.38	6.22	92.96
High threshold (17 U/ml)	-	5.13	92.08	4.44	93.11
ETAR (threshold 10 U/ml)	0.56 (0.47-0.66)	30.77	57.46	4.94	92.04
C3aR	0.51 (0.42–0.61)	25.64	83.06	9.80	93.96
C35aR	0.56 (0.46-0.66)	53.85	59.67	8.75	94.74

AT1R, angiotensin II receptor type I; C3aR, complement C34a receptor; C5aR, complement C5a receptor; CI, confidence interval; ETAR, endothelin receptor 1 receptor type A; HLA, human leukocyte antigen; ROC-AUC, area under the receiver operating characteristic curve.

Positive thresholds are based on the literature for AT1R and ETAR antibodies and calculated with the Youden index for C3aR and C5aR antibodies (as there is no threshold published in previous studies for these last 2 biomarkers).



Figure 6 | Association of emerging blood biomarkers and standard of care parameters at the time of biopsy with kidney allograft rejection. This forest plot shows the univariate association of blood mRNA gene expression, non-HLA antibodies, and routinely collected parameters with kidney allograft rejection. The numbers correspond to odds ratios (ORs) and the horizontal lines correspond to 95% confidence intervals of the respective odds ratios. Statistically significant associations are represented by black squares (* $P \le 0.05$, ** $P \le 0.01$, *** $P \le 0.001$), and nonsignificant associations are represented by transparent circles. AT1R_Ab, antibodies against angiotensin II receptor type 1; C3aR_Ab, antibodies against complement C3a receptor; DSA, donor-specific antibody; ECD, expanded criteria donor; eGFR, estimated glomerular filtration rate; ETAR_Ab, antibodies against endothelin 1 receptor type 1; iDSA, immunodominant DSA; MFI, mean fluorescence intensity; non-HLA, non-human leukocyte antigen; SCD, standard criteria donor; UPCR, urine protein-to-creatinine ratio.

Banff 2022 classification;⁴⁵ and (iii) the use of elastic net regularized regressions, which allow us to generate the optimal combinations of blood biomarkers to increase predictive performances of the diagnostic models, showing that even combined optimally, they did not have an incremental value beyond standard of care parameters.

Finally, our data show that unselected prospective cohort studies are a critical step to validate the clinical utility of monitoring biomarkers, before moving toward implementation in clinical care.²⁵ Indeed, unselected studies allow to capture natural disease prevalence and correctly assess biomarkers in settings that mimic real life and capture the full



Figure 7 | Comparison of the performances of the predictive models of kidney allograft rejection developed using blood biomarkers, routine parameters, or all parameters. Discriminative abilities of the standard of care (SOC) model, biomarker model, and integrative model (SOC + biomarkers) to detect kidney allograft rejection, assessed through the calculation of the area under the receiver operating characteristic curve (ROC-AUC). To note, 6 blood mRNA biomarkers (*AKR1C3, CTLA4, FOXP3, MZB1, POU2AF1*, and *TCL1A*) were excluded from these analyses because of a rate of missing values >30% mainly secondary to technical issues in routine monitoring. CI, confidence interval.

spectrum of kidney allograft phenotypes.²⁵ For instance, the blood gene expression assay, named kidney Solid Organ Response Test (kSORT), showed promising diagnostic performances in case-control proof of concept studies, but then failed to be clinically validated in an unselected cohort.^{54,55} Still, these results could be used for health technology assessment and integrated into de novo decision models in a broad spectrum of clinical settings and national contexts.^{56,57} However, conducting such studies is challenging⁵⁸ and requires, for instance, for this work joint efforts of a multisite kidney transplant network including a range of researchers, immunologists, nephrologists, pathologists, pharmacologists, statisticians, public health authorities, and industrial partners. Hence, because of time and financial constraints, biomarkers are often initially assessed in retrospective selected casecontrol studies, which do not reflect natural disease prevalence and are thus at high risk of bias.^{26,27} Here, our robust design allowed us to demonstrate that the panel of blood mRNAs and non-HLA antibodies assessed in this study were not significantly associated with allograft rejection in the first year after transplantation, and displayed a multicollinearity reflecting that they share a significant biological redundancy.^{25,59} Indeed, most of these biomarkers have been described in relation to immune-related mechanisms. B cell–related genes were associated with allograft tolerance and others,^{57,58} such as *FOXP3* and *CTLA4*, are well recognized in immune tolerance.⁶ This shows that these genes are modulated in specific immune-related situations and may imply specific contexts of use, such as *AKR1C3* and *TCL1A*, in operational tolerance and subclinical AMR.^{7,27} This suggests that these results may not be generalized for all events and at any time after transplantation.

Our study has limitations. First, we followed patients only during the first year after transplantation; hence, our results are valid in this context of use but are not generalizable beyond this period. In contrast, in this study, most biopsies were analyzed early after transplantation (median time of 4.1 months after transplantation). Because mRNA parameters measured in total blood are sensitive to cell composition, their levels may be altered by perioperative treatments and induction therapies. Secondly, although our cohort was multicentric, all the patients were European, and we did not have access to patient ethnicity data. Thirdly, the cumulative incidence of the primary outcome (allograft rejection) was

Table 6 Sensiti	vity analyses	in different	clinical	scenarios
and subpopulat	ions			

Clinical scenarios and subpopulations	Biopsies, n	Events, n	ROC-AUC (95% CI)
In stable allografts	625	18	
Biomarkers alone			0.58 (0.47–0.69)
SOC alone			0.70 (0.56-0.83)
SOC + biomarkers (integrative model)			0.72 (0.59–0.84)
In unstable allografts	187	34	
Biomarkers alone			0.67 (0.56-0.78)
SOC alone			0.77 (0.69–0.85)
SOC + biomarkers (integrative model)			0.69 (0.60–0.78)
In biopsies with AMR	784	24	
Biomarkers alone			0.66 (0.54–0.77)
SOC alone			0.78 (0.67–0.89)
SOC + biomarkers (integrative model)			0.77 (0.68–0.87)
In biopsies with TCMR	785	25	
Biomarkers alone			0.72 (0.62–0.82)
SOC alone			0.85 (0.77–0.93)
SOC + biomarkers (integrative model)			0.85 (0.77–0.93)
In biopsies <3 months after transplantation	200	23	
Biomarkers alone			0.71 (0.59–0.83)
SOC alone			0.80 (0.69–0.90)
SOC + biomarkers (integrative model)			0.77 (0.67–0.86)
In biopsies \geq 3 months after transplantation	612	29	
Biomarkers alone			0.62 (0.51–0.72)
SOC alone			0.81 (0.71–0.90)
SOC + biomarkers (integrative model)			0.81 (0.72–0.90)
CNI therapy at the time of biopsy	740	45	
Biomarkers alone			0.69 (0.62–0.77)
SOC alone			0.83 (0.76–0.90)
SOC + biomarkers (integrative model)			0.83 (0.77–0.89)
No CNI therapy at the time of biopsy	44	4	
Biomarkers alone			0.61 (0.30–0.93)
SOC alone			0.68 (0.39–0.97)
SOC + biomarkers (integrative model)			0.68 (0.36–0.99)
In French centers ^a	629	30	
Biomarkers alone			0.70 (0.59–0.80)
SOC alone			0.82 (0.73–0.91)
SOC + biomarkers (integrative model)			0.82 (0.73–0.90)
In non-French centers ^b	161	22	
Biomarkers alone			0.62 (0.49–0.75)
SOC alone			0.77 (0.66–0.88)
SOC + biomarkers (integrative model)			0.75 (0.64–0.86)

(Continued)

Table 6 (Continued)

Clinical scenarios and subpopulations	Biopsies, n	Events, n	ROC-AUC (95% CI)
Allograft rejection defined according to the international Banff 2022 classification ⁴⁵	812	61	
Biomarkers alone			0.55 (0.46-0.66)
SOC alone			0.80 (0.68-0.92)
SOC + biomarkers (integrative model)			0.80 (0.68–0.91)

AMR, antibody-mediated rejection; CI, confidence interval; CNI, calcineurin inhibitor; EU-TRAIN, European TRAnsplantation and INnovation; ROC-AUC, area under the receiver operating characteristic curve; SOC, standard of care; TCMR, T cell-mediated rejection.

^aParis-Saint-Louis, Paris-Necker, Nantes, and Paris-Kremlin-Bicêtre.

^bBarcelona-Bellvitge, Barcelona-Vall d'Hebron, Berlin-Charité Mitte, Berlin-Charité Virchow, and Geneva.

We applied the locked parameters of the EU-TRAIN biomarker model (developed using only blood biomarkers), SOC model (developed using only routine parameters), and integrative model (developed using blood biomarkers and routine parameters) in different scenarios and subpopulations to assess their discriminative abilities to detect kidney allograft rejection in these settings. Allograft instability was defined as a 20% increase in serum creatinine and/or a worsening or appearance of a urine protein-to-creatinine ratio \geq 0.5 g/g at the time of biopsy.

low (8.1% at 1 year after transplantation). However, it was in line with epidemiologic data gathered in the modern era of immunosuppression,⁵² and we included the target number patients, calculated a priori, to have a sufficient statistical power. Fourthly, although there was not any statistical tendency for an additional value of the EU-TRAIN blood biomarkers in the various exploratory sensitivity analyses, this study was not specifically designed to address these outcomes. Finally, our analysis was restricted to 23 candidate blood biomarkers, but we constituted a large biobank of blood samples, which will be stored for 30 years for potential future biomarker validation studies. Furthermore, we experienced a high number of missing values for some genes (AKR1C3, CTLA4, FOXP3, MZB1, POU2AF1, and TCL1A), which may logically be related to the low level of expressing cells in total blood, such as regulatory T cells for FOXP3. For these lowly expressed genes, the use of conventional quantitative polymerase chain reaction with enzymatic amplification may have reduced the number of missing values and increased measure precision for these genes close to the limit of detection with the multiplex gene expression method used. None of these 6 genes showed statistical associations with allograft rejection in univariate analysis.

In conclusion, we demonstrate that the 23 EU-TRAIN blood biomarkers of gene expression profiling and non-HLA antibodies targeting endothelial antigens assessed in this large prospective multicenter study were not significantly associated with kidney allograft rejection. This questions their clinical utility for monitoring rejection in adult kidney transplant recipients during the first year after transplantation and justifies their evaluation in specific contexts of use. The EU-TRAIN unselected and multimodal design paves the way for future studies assessing the relevance of biomarkers for clinical decision-making in kidney transplantation.

DISCLOSURE

RD and SB have ownership interests in the BioMAdvanced Diagnostics company, which is not involved in the present research. HH is chief executive officer of CellTrend. AL holds shares in Predict4Health, a software company that is not involved in the present research. All the other authors declared no competing interests.

DATA STATEMENT

Data are available from the corresponding author on reasonable request, including standards for General Data Protection Regulation and approval by the Ethics Committee of the EU-TRAIN's consortium.

ACKNOWLEDGMENTS

The authors thank Thiên-Ly Pham and Dahlia Tsakiropoulos for their contribution in the management of the project; Laurence Letertre for her help in setting up and developing the project; Marta Sablik for designing Figure 1; and the clinical research assistants Marie Bouclier, Francine Tacafred, Julie Obert, Solène Janique, Aliaume Le Pape, Maxime Bentoumi-Loaec, Claudia Fink, Claudia Beck, Uwe Behrens, Fernando Ardila Valle, Sara Gestrudis Horna Quintana, Mireia Sanllorente, Jocelyne Chabert, Elodie Von Dach, Emmanuelle Lelong-Favre, and Fatma Bair for their contribution in the data acquisition. The authors also thank Blaise Robin, Jessy Dagobert, and Fariza Mezine for their technical assistance, and the Biological Resource Centre for Biobanking (CHU Nantes, Nantes Université, Centre de ressources biologiques [BRIF: BB-0033-00040]).

Additional information

This paper is dedicated to our dear friend Duska Dragun, who has passed away too early.

FUNDING

The EU-TRAIN study was funded by the European Union's Horizon 2020 research and innovation program under grant agreement no. 754995. VG received grants from the French-Speaking Society of Transplantation and the French Foundation for Medical Research. RD was supported by the EU-TRAIN study and the ANR project KTD-innov (ANR-17-RHUS-0010) thanks to the French Government financial support managed by the National Research Agency (ANR) within "Investments into the Future" program. OA received a grant from the Foundation Bettencourt-Schueller.

AUTHOR CONTRIBUTIONS

AL, CLef, OB, DD, and KB designed the EU-TRAIN study. VG, RD, RAC, MRac, AP, ME, MRay, OA, DB, FG, EV, SY, JD, HH, JLT, CRL, MZ, EP, HLM, TVHN, FM, TB, JV, CLef, DD, VP, LP, MG, PAG, SB, EC, FH, KB, OB, AL and CLef participated in the building of the EU-TRAIN cohort and contributed to data acquisition. VG, MRac, MRay, OA, AL, and CLef performed data analysis. VG, RD, RAC, MRac, MRay, OA, AL, and CLef performed data interpretation. VG, MRac, AL, and CLef wrote the first draft of the manuscript. All authors revised and critically reviewed the manuscript.

Supplementary material is available online at www.kidneyinternational.org.

REFERENCES

- Mayrdorfer M, Liefeldt L, Wu K, et al. Exploring the complexity of deathcensored kidney allograft failure. J Am Soc Nephrol. 2021;32:1513–1526.
- 2. Muduma G, Odeyemi I, Smith-Palmer J, Pollock RF. Review of the clinical and economic burden of antibody-mediated rejection in renal transplant recipients. *Adv Ther.* 2016;33:345–356.
- 3. Loupy A, Haas M, Roufosse C, et al. The Banff 2019 Kidney Meeting Report (I): updates on and clarification of criteria for T cell- and antibodymediated rejection. *Am J Transplant*. 2020;20:2318–2331.
- 4. Loupy A, Mengel M, Haas M. Thirty years of the International Banff Classification for Allograft Pathology: the past, present, and future of kidney transplant diagnostics. *Kidney Int*. 2022;101:678–691.
- Bu L, Gupta G, Pai A, et al. Clinical outcomes from the Assessing Donorderived cell-free DNA Monitoring Insights of kidney Allografts with Longitudinal surveillance (ADMIRAL) study. *Kidney Int.* 2022;101:793–803.
- Danger R, Chesneau M, Paul C, et al. A composite score associated with spontaneous operational tolerance in kidney transplant recipients. *Kidney Int.* 2017;91:1473–1481.
- Danger R, Le Berre L, Cadoux M, et al. Subclinical rejection-free diagnostic after kidney transplantation using blood gene expression. *Kidney Int.* 2023;103:1167–1179.
- Rabant M, Amrouche L, Lebreton X, et al. Urinary C-X-C Motif Chemokine 10 Independently Improves the Noninvasive Diagnosis of Antibody-Mediated Kidney Allograft Rejection. J Am Soc Nephrol. 2015;26:2840–2851.
- 9. Van Loon E, Tinel C, de Loor H, et al. Automated urinary chemokine assays for noninvasive detection of kidney transplant rejection: a prospective cohort study. *Am J Kidney Dis.* 2024;83:467–476.
- Suthanthiran M, Schwartz JE, Ding R, et al. Urinary-cell mRNA profile and acute cellular rejection in kidney allografts. N Engl J Med. 2013;369:20–31.
- Lubetzky ML, Salinas T, Schwartz JE, Suthanthiran M. Urinary cell mRNA profiles predictive of human kidney allograft status. *Clin J Am Soc Nephrol.* 2021;16:1565–1577.
- 12. El Fekih R, Hurley J, Tadigotla V, et al. Discovery and validation of a urinary exosome mRNA signature for the diagnosis of human kidney transplant rejection. *J Am Soc Nephrol.* 2021;32:994–1004.
- Fichtner A, Süsal C, Höcker B, et al. Association of non-HLA antibodies against endothelial targets and donor-specific HLA antibodies with antibody-mediated rejection and graft function in pediatric kidney transplant recipients. *Pediatr Nephrol.* 2021;36:2473–2484.
- Philogene MC, Bagnasco S, Kraus ES, et al. Anti-angiotensin II type 1 receptor and anti-endothelial cell antibodies: a cross-sectional analysis of pathological findings in allograft biopsies. *Transplantation*. 2017;101: 608–615.
- Ashton-Chess J, Giral M, Mengel M, et al. Tribbles-1 as a novel biomarker of chronic antibody-mediated rejection. J Am Soc Nephrol. 2008;19:1116– 1127.
- Viklicky O, Krystufkova E, Brabcova I, et al. B-cell-related biomarkers of tolerance are up-regulated in rejection-free kidney transplant recipients. *Transplantation*. 2013;95:148–154.
- 17. Heidt S, Vergunst M, Anholts JDH, et al. B cell markers of operational tolerance can discriminate acute kidney allograft rejection from stable graft function. *Transplantation*. 2015;99:1058–1064.
- Pallier A, Hillion S, Danger R, et al. Patients with drug-free long-term graft function display increased numbers of peripheral B cells with a memory and inhibitory phenotype. *Kidney Int*. 2010;78:503–513.
- **19.** Brouard S, Mansfield E, Braud C, et al. Identification of a peripheral blood transcriptional biomarker panel associated with operational renal allograft tolerance. *Proc Natl Acad Sci U S A*. 2007;104:15448–15453.
- Louis K, Fadakar P, Macedo C, et al. Concomitant loss of regulatory T and B cells is a distinguishing immune feature of antibody-mediated rejection in kidney transplantation. *Kidney Int*. 2022;101:1003–1016.
- 21. Lefaucheur C, Louis K, Philippe A, et al. The emerging field of non-human leukocyte antigen antibodies in transplant medicine and beyond. *Kidney Int*. 2021;100:787–798.
- 22. Dragun D, Müller DN, Bräsen JH, et al. Angiotensin II type 1-receptor activating antibodies in renal-allograft rejection. *N Engl J Med.* 2005;352: 558–569.
- 23. Lefaucheur C, Viglietti D, Bouatou Y, et al. Non-HLA agonistic antiangiotensin II type 1 receptor antibodies induce a distinctive phenotype of antibody-mediated rejection in kidney transplant recipients. *Kidney Int.* 2019;96:189–201.

ARTICLE IN PRESS

clinical investigation

- 24. Pearl MH, Chen L, ElChaki R, et al. Endothelin type A receptor antibodies are associated with angiotensin II type 1 receptor antibodies, vascular inflammation, and decline in renal function in pediatric kidney transplantation. *Kidney Int Rep.* 2020;5:1925–1936.
- 25. Naesens M, Anglicheau D. Precision transplant medicine: biomarkers to the rescue. J Am Soc Nephrol. 2018;29:24–34.
- 26. Ioannidis JPA, Bossuyt PMM. Waste, leaks, and failures in the biomarker pipeline. *Clin Chem.* 2017;63:963–972.
- Raynaud M, Al-Awadhi S, Louis K, et al. Prognostic biomarkers in kidney transplantation: a systematic review and critical appraisal. J Am Soc Nephrol. 2024;35:177–188.
- Jamshaid F, Froghi S, Di Cocco P, Dor FJ. Novel non-invasive biomarkers diagnostic of acute rejection in renal transplant recipients: a systematic review. *Int J Clin Pract*. Published online July 16, 2018. https://doi.org/10. 1111/ijcp.13220
- 29. Menon MC, Murphy B, Heeger PS. Moving biomarkers toward clinical implementation in kidney transplantation. *J Am Soc Nephrol.* 2017;28: 735–747.
- Vandenbroucke JP, von Elm E, Altman DG, et al. Strengthening the Reporting of Observational Studies in Epidemiology (STROBE): explanation and elaboration. *Epidemiology*. 2007;18:805–835.
- Cohen JF, Korevaar DA, Altman DG, et al. STARD 2015 guidelines for reporting diagnostic accuracy studies: explanation and elaboration. *BMJ Open*. 2016;6:e012799.
- **32.** Heidari S, Babor TF, De Castro P, et al. Sex and Gender Equity in Research: rationale for the SAGER guidelines and recommended use. *Res Integr Peer Rev.* 2016;1:2.
- **33.** Tambur AR, Campbell P, Claas FH, et al. Sensitization in Transplantation: Assessment of Risk (STAR) 2017 Working Group Meeting Report. *Am J Transplant*. 2018;18:1604–1614.
- **34.** Yoo D, Goutaudier V, Divard G, et al. An automated histological classification system for precision diagnostics of kidney allografts. *Nat Med.* 2023;29:1211–1220.
- **35.** Riley RD, Ensor J, Snell KIE, et al. Calculating the sample size required for developing a clinical prediction model. *BMJ*. 2020;368:m441.
- **36.** Kohavi R. A study of cross-validation and bootstrap for accuracy estimation and model selection. Proceedings of the 14th International Joint Conference on Artificial Intelligence. 1995:1137–1145.
- Sperrin M, Riley RD, Collins GS, Martin GP. Targeted validation: validating clinical prediction models in their intended population and setting. *Diagn Progn Res.* 2022;6:24.
- DeLong ER, DeLong DM, Clarke-Pearson DL. Comparing the areas under two or more correlated receiver operating characteristic curves: a nonparametric approach. *Biometrics*. 1988;44:837–845.
- **39.** Kuhn M. Building predictive models in R using the caret package. *J Stat Softw.* 2008;28:1–26.
- **40.** Collins GS, Reitsma JB, Altman DG, Moons KGM. Transparent Reporting of a multivariable prediction model for Individual Prognosis or Diagnosis (TRIPOD): the TRIPOD statement. *Ann Intern Med.* 2015;162:55–63.
- 41. Little RJ, D'Agostino R, Cohen ML, et al. The prevention and treatment of missing data in clinical trials. *N Engl J Med*. 2012;367:1355–1360.
- **42.** Stekhoven DJ, Bühlmann P. MissForest—non-parametric missing value imputation for mixed-type data. *Bioinformatics*. 2012;28:112–118.

- 43. Tang F, Ishwaran H. Random forest missing data algorithms. *Stat Anal Data Min.* 2017;10:363–377.
- **44.** Waljee AK, Mukherjee A, Singal AG, et al. Comparison of imputation methods for missing laboratory data in medicine. *BMJ Open.* 2013;3: e002847.
- Naesens M, Roufosse C, Haas M, et al. The Banff 2022 Kidney Meeting Report: reappraisal of microvascular inflammation and the role of biopsybased transcript diagnostics. *Am J Transplant*. 2024;24:338–349.
- **46.** Philippe A, Arns W, Ditt V, et al. Impact of everolimus plus calcineurin inhibitor on formation of non-HLA antibodies and graft outcomes in kidney transplant recipients: 12-month results from the ATHENA substudy. *Front Transplant*. 2023;2:1273890.
- Lefaucheur C, Louis K, Morris AB, et al. Clinical recommendations for posttransplant assessment of anti-HLA (human leukocyte antigen) donor-specific antibodies: a Sensitization in Transplantation: Assessment of Risk consensus document. *Am J Transplant*. 2023;23: 115–132.
- **48.** Oweira H, Ramouz A, Ghamarnejad O, et al. Risk factors of rejection in renal transplant recipients: a narrative review. *J Clin Med.* 2022;11: 1392.
- 49. Hart A, Singh D, Brown SJ, et al. Incidence, risk factors, treatment, and consequences of antibody-mediated kidney transplant rejection: a systematic review. *Clin Transplant*. 2021;35:e14320.
- Foroutan F, Friesen EL, Clark KE, et al. Risk factors for 1-year graft loss after kidney transplantation: systematic review and meta-analysis. *Clin J Am Soc Nephrol.* 2019;14:1642–1650.
- Lemoine M, Titeca Beauport D, Lobbedez T, et al. Risk factors for early graft failure and death after kidney transplantation in recipients older than 70 years. *Kidney Int Rep.* 2019;4:656–666.
- 52. Matas AJ, Smith JM, Skeans MA, et al. OPTN/SRTR 2011 Annual Data Report: kidney. *Am J Transplant*. 2013;13(suppl 1):11–46.
- Steinberg DM, Fine J, Chappell R. Sample size for positive and negative predictive value in diagnostic research using case–control designs. *Biostatistics (Oxford, England)*. 2009;10:94.
- Roedder S, Sigdel T, Salomonis N, et al. The kSORT assay to detect renal transplant patients at high risk for acute rejection: results of the multicenter AART study. *PLoS Med.* 2014;11:e1001759.
- 55. Van Loon E, Giral M, Anglicheau D, et al. Diagnostic performance of kSORT, a blood-based mRNA assay for noninvasive detection of rejection after kidney transplantation: a retrospective multicenter cohort study. *Am J Transplant*. 2021;21:740–750.
- Girardin FR, Cohen K, Schwenkglenks M, Durand-Zaleski I. Editorial: Pharmacoeconomics in the era of health technology assessment and outcomes research to prioritize resource use, innovation and investment. *Front Pharmacol.* 2023;14:1210002.
- Girardin FR, Nicolet A, Bestard O, et al. Immunosuppressant drugs and quality-of-life outcomes in kidney transplant recipients: an international cohort study (EU-TRAIN). *Front Pharmacol.* 2023;14:1040584.
- 58. Sarwal MM, Naesens M. Urine trumps the protocol biopsy for subclinical rejection surveillance. *Kidney Int*. 2023;104:432–439.
- 59. Christensen NJ, Demharter S, Machado M, et al. Identifying interactions in omics data for clinical biomarker discovery using symbolic regression. *Bioinformatics*. 2022;38:3749–3758.